Author’s response to reviews

Title: Immunohistochemical determination of the miR-1290 target arylamine N-acetyltransferase 1 (NAT1) as a prognostic biomarker in breast cancer

Authors:

Yumi Endo (yendo424@med.nagoya-cu.ac.jp)
Hiroko Yamashita (hirokoy@huhp.hokudai.ac.jp)
Satoru Takahashi (sattak@med.nagoya-cu.ac.jp)
Shinya Sato (satopin@med.nagoya-cu.ac.jp)
Nobuyasu Yoshimoto (paris@s4.dion.ne.jp)
Tomoko Asano (rinkomaneko@hotmail.co.jp)
Yukari Hato (yukarinhato@gmail.com)
Yu Doung (donyufish@yahoo.co.jp)
Yoshitaka Fujii (yosfuji@med.nagoya-cu.ac.jp)
Tatsuya Toyama (t.toyama@med.nagoya-cu.ac.jp)

Version: 2
Date: 3 December 2014

Author’s response to reviews: see over
Dr. Emily Crow,
BMC Cancer

Re: MS: 8573821881344170

Title: Immunohistochemical determination of the miR-1290 target arylamine N'-acetyltransferase 1 (NAT1) as a prognostic biomarker in breast cancer

Authors: Yumi Endo, Hiroko Yamashita, Satoru Takahashi, Shinya Sato, Nobuyasu Yoshimoto, Tomoko Asano, Yukari Hato, Yu Dong, Yoshitaka Fujii, Tatsuya Toyama

Dear Dr. Crow,

We are most grateful to the editors and the reviewers for their critical review and helpful comments on the manuscript.

You requested that the name of the ethics committee that approved our study should be included. Therefore, we have changed the sentences from “The study protocol was approved by the institutional review board” to “This protocol was approved by the Institutional Review Board of Nagoya City University Graduate School of Medical Sciences” in the section of “Methods” according to the request.

This change has been highlighted in the revised manuscript.

We hope that our responses are adequate.

Thank you for your attention to our manuscript.
Sincerely,

Tatsuya Toyama

Department of Oncology, Immunology and Surgery,
Nagoya City University Graduate School of Medical Sciences
1 Kawasumi, Mizuho-cho, Mizuho-ku, Nagoya 467-8601, Japan
Phone: +81-52-853-8231 Fax: +81-52-853-6440
Email: t.toyama@med.nagoya-cu.ac.j